Author's response to reviews

Title: Drug discovery in ophthalmology: past success, present challenges, and future opportunities

Authors:

Nicholas JD Gower (nicholas.gower@nhs.net)
Robert J Barry (R.J.Barry@bham.ac.uk)
Matthew R Edmunds (M.R.Edmunds@bham.ac.uk)
Lucy C Titcomb (lucy.titcomb@nhs.net)
Alastair K Denniston (A.Denniston@bham.ac.uk)

Version: 5 Date: 5 January 2016

Author's response to reviews: see over
5th January 2016

BMC Ophthalmology

Dear Editor

Re: Drug discovery in ophthalmology: past success, present challenges, and future opportunities

Thank you for your email dated 14th December 2015. We have amended the wording of the "Authors' contributions" so that each author is listed individually. We look forward to hearing from you in the near future.

Yours sincerely,

Alastair Denniston
Hon Senior Lecturer in Ophthalmology/ Consultant Ophthalmologist